Author's response to reviews

Title: Sick-leave track record and other potential predictors of a disability pension
A population based study of 8,218 men and women followed for 16 years

Authors:

Thorne Wallman (thorne.wallman@pubcare.uu.se)
Hans Wedel (hans.wedel@biostat.se)
Edward Palmer (edward.palmer@forsakringskassan.se)
Annika Rosengren (annika.rosengren@gu.se)
Saga Johansson (saga.johansson@astrazeneca.com)
Henry Eriksson (henryeriksson@swipnet.se)
Kurt Svärdsudd (kurt.svardsudd@pubcare.uu.se)

Version: 6 Date: 4 April 2009

Author's response to reviews:

The requested changes in Table 1 and the reference list have been performed. In addition, the text has been proof read and a number of minor text adjustments have been made.